\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at108a.txt}{\rawdatadir/at108a.txt}~Freiwald (1962)~\cite{h_frei62}
\W\label{at108a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at108a~\protect\cite[Freiwald (1962)]{h_frei62}\label{at108a}}\\
\T\input{\datadir/at108a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & .7-4.1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at108b.txt}{\rawdatadir/at108b.txt}~Kogarko (1965)~\cite{s_koga65}
\W\label{at108b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at108b~\protect\cite[Kogarko (1965)]{s_koga65}\label{at108b}}\\
\T\input{\datadir/at108b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & spherical, high explosive & Oxidizer: & Air \\
Initial Pressure: & 101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at82c.txt}{\rawdatadir/at82c.txt}~Lee (1977)~\cite{j_lee77}
\W\label{at82c}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at82c~\protect\cite[Lee (1977)]{j_lee77}\label{at82c}}\\
\T\input{\datadir/at82c.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & spherical & Oxidizer: & O2 \\
Initial Pressure: & 4.1-12.2 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at82b.txt}{\rawdatadir/at82b.txt}~Lee (1977)~\cite{j_lee77}
\W\label{at82b}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at82b~\protect\cite[Lee (1977)]{j_lee77}\label{at82b}}\\
\T\input{\datadir/at82b.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & cylindrical & Oxidizer: & O2 \\
Initial Pressure: & 6.1-40.5 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at44a.txt}{\rawdatadir/at44a.txt}~Matsui (1979)~\cite{h_mats79}
\W\label{at44a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{cc}
\T\caption{at44a~\protect\cite[Matsui (1979)]{h_mats79}\label{at44a}}\\
\T\input{\datadir/at44a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & spherical & Oxidizer: & O2 \\
Initial Pressure: &  kPa & Diluent: &  \\
Initial Temperature: &  K & Equivalence Ratio: & 0.7 - 2.1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at54a.txt}{\rawdatadir/at54a.txt}~Vasil'ev (1982)~\cite{a_vasi82}
\W\label{at54a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at54a~\protect\cite[Vasil'ev (1982)]{a_vasi82}\label{at54a}}\\
\T\input{\datadir/at54a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & cylindrical & Oxidizer: & O2 \\
Initial Pressure: & 2.7-5.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

\W\begin{center}
\W\begin{table}[!ht]
\W\xlink{at105a.txt}{\rawdatadir/at105a.txt}~Vasil'ev (1980)~\cite{a_vasi80}
\W\label{at105a}
\W\end{table}
\W\end{center}
\T\begin{longtable}{ccc}
\T\caption{at105a~\protect\cite[Vasil'ev (1980)]{a_vasi80}\label{at105a}}\\
\T\input{\datadir/at105a.txt}
\T\end{longtable}

\begin{center}
\begin{tabular}{rlrl}
Category: & critical energy & Fuel: & C2H2 \\
Sub-Category: & cylindrical, high explosive & Oxidizer: & Air \\
Initial Pressure: & 30.4-101.3 kPa & Diluent: &  \\
Initial Temperature: & 293 K & Equivalence Ratio: & 1 \\
\end{tabular}
\end{center}

